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KINGSTON ASH RECOVERY PROJECT

WEEKLY REPORT
May 18 - 24, 2009

HIGHLIGHTS

The ash stilling pond dredging was initiated and approximately 3,000 cubic yards removed.

Held a briefing on the root cause analysis for the TDEC technical team on May 21% in Nashville.
About 18,000 cubic yards of river ash was removed from the Emory River.

Received EPA approval on 5 minor work plans; rail spur, rail car offloading, barge area offload, ash
movement east of Dike 2, and movement of yellow dredge.

Submitted draft safety and health plan and time critical removal action memorandum to EPA.

o Work progressing on schedule on the rail spur construction. Nearly 800 feet of rail and ties have been
positioned for installation.

INFRASTRUCTURE/ASH MANAGEMENT

Engineering
e Car wash design and installation has been put on hold — determination of location is pending.

e Awaiting survey results from TVA to transmit to CTI (Utilities Engineer) so they can complete the
engineering for utilities replacement along Swan Pond Road and other locations.

e Preparing sketch to widen the radius at the intersection of Swan Pond Road and Swan Pond Circle.

e Preparing sketch to eliminate the S curve in the middle entrance road to the facility.

e Prepared the Railroad Spur Construction Work Plan for review/approval by EPA.

Roads

e The Civil Projects Group (CPG) completed the installation of the ramp at Dike C.

¢ Completed filling the ditch on the main access haul road for the relocation of the wheel wash and widening
the road to accommodate the new rail spur.

Utilities
¢ No construction activity at this time.

Water Management

¢ Completed excavating ash and contouring the base of Sediment Pond 2A and preparing to install the
stand pipe. Have hauled approximately 76,850 cubic yards of ash from the Sediment Ponds and area
east of Dike 2 to the Ash Processing Area.

e Determination made and approval received to install a fabric baffle in Sediment Pond 2A in lieu of the
stone baffle as designed. The stone baffle in Sediment Pond 1A will remain as installed.

e Continued construction of the clean water ditches — from the north embayment to Dike 2.

e Continued installing the temporary rock road in the north embayment for accessibility to construct the berm
and move the ash from the northern most point of the embayment.

e Continued excavating the discharge outlets east of Dike 2 for the clean water lines and Sediment Pond
1A

e Constructed road off Dike 2 and began hauling clay from the clean water ditch excavation to the gypsum
pond.

o Completed repair of the slough area and installation of the pipe west of Dike 2.

¢ Continued installation of check dams and re-contouring of area for effective water management/drainage.

o Performed Storm Water Pollution Prevention Plan (SWPPP) inspections as required.

Dust/Erosion Control

o 14.8 acres of Flexterra applied this week.

o 321 acres of Flexterra applied since January 31, 2009.

e Continuing operating water trucks on roads as needed for dust control

¢ Working with TVA personnel on additional short and long term remedies for dust control — seeding,
alternatives to Flexterra.

e Revising the Fugitive Dust Control and Suppression for submittal to the EPA.
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ASH DREDGING & PROCESSING

Environmental Data Collection (TSS)

e Continued daily TSS monitoring at KIF NPDES 001. The May monthly average through Thursday, May
21,2009 is 10.23 mg/L.

e Dredge plume monitoring was performed Monday through Saturday, May 18 - May 23, 2009. There is no
elevated turbidity in the river.

Dredging
o Approximately 17,900 cubic yards of material was dredged from the Emory River between May 18" and

May 24™.

e Dredging cumulative to date volume: 156,000 cubic yards.

e Mechanical issues with the clutch on Luzon persisted last week.

¢ Ash Stilling Pond dredging was conducted as a 12-hour dayshift activity all week with an estimated
material volume of 3,100 cubic yards.

e Prepared work plans for the relocation of the TVA (yellow) dredge and installation of a barge off-loading
station. The work plans were approved by EPA.

e Prepared the TVA dredge for relocation from the settling pond to the Emory River.

o Established a north barge off-loading station on the northern end of Segment 1 to support additional debris
removal.

e  Submitted the Time Critical Action Memorandum for dredging on May 22",

e Crews were deployed on May 22" to perform sampling of ash/sediment for visual examination across
several sections of the Emory and Clinch Rivers. This is to determine if upstream sediments have entered
the work zone related to the movement of ash caused by the elevated flow event of May 4™.

Ash Processing:

e The Ash Processing and Storage area contains approximately 240,000 cubic yards of fly ash. A portion of
the stockpile is from dredging and recovery operations and the remainder is from the embayment area
(material excavated from the installation of stormwater ponds and clean/dirty water ditches).

ASH DISPOSITION

e Remaining leaks in the settlement plate water-line were repaired on Friday, May 22nd. Final hydrostatic
pressure test was performed on Friday and was satisfactory in maintaining pressure.

e Unloading operations for the test rail cars that did not leave the KIF site will begin on Tuesday, May 26",
A work plan for this activity was submitted and approved by EPA.

o Best and final offers from Loading and Disposal Contractors were due May 22" _ planned contract award
is scheduled for Friday, May 29".

CENOSPHERE RECOVERY

e Continue to remove debris and cenospheres generated from the heavy rain on Sunday, May 3".

e Continued to use the barge-mounted excavator to remove large debris.

e Areas of focus have been maintaining the booms and intake channel in Division A and B, the waterfront
park in Kingston and outreach areas down river. The area down river referred to as Lake Harbor has been
completed.

¢ Individual outreach calls have been addressed as equipment frees up.

e The river and shoreline was in good shape for the holiday weekend.

e Boats and equipment were moved back to the staging area for the long weekend.
e A safe navigation channel marked by buoys was installed.

SKIMMER WALL

e Dredging in Skimmer Wall area to uncover disabled wall is continuing.
e River OES divers will be surveying dredging operations in the location of the disabled wall on Tuesday,
May 26" and will report resullts.

DIKE REINFORCEMENT

e Stantec has reduced the number of borings from 74 to 63. Based on fairly uniform site conditions
encountered within areas of the boring program Stantec has elected to delete the following borings from
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the Ash Pond stability drilling program: STN-70, STN-39, STN-68, STN-67, STN-33, STN-58, and STN-
57. Based on access restriction within the failed ash, the following borings from the Ash Pond stability
drilling program have been deleted: STN-1, STN-7, and STN-13. Based on the proximity of a previously
drilled AECOM conventional boring, STN-44 has been deleted.

e Stantec has completed 51 of the 63 proposed borings for analyzing dike stability. Stantec made a
presentation on the progress of the dike stability at KIF to TDEC on May 21% as part of the Root Cause
Report. The remaining locations should be completed by mid June.

e Reached agreement with the regulators on a path forward for removing the trees and placing riprap along
Dike C. Work on the riprap repair should be completed by the end of next week.

e Completed construction of the new access ramp at Dike C.

ROUTINE MONITORING

Surface Water Sampling

¢ Routine surface water fixed station monitoring occurred on Monday, Wednesday, and Friday, May 18™,
20", and 22", respectively.

e The monthly sampling and toxicity testing of the KIF Stilling Pond discharge was initiated on Sunday, May
24" with sample collection to continue through Friday, May 29".

Air Sampling

e Fixed-location and real-time monitoring continued.

¢ No exceedences of standards were observed from real-time measurements.

o TDEC conducted a follow-up audit of the fixed-station samplers on May 19™. The audit report has not
been received.

Data Management

¢ Electronic data packages continue to be received from the metals re-analysis of surface water samples
collected in January and February 2009.

e Data validation continues for electronic data packages received from the laboratories. As they are
received, the verified packages are being posted to the password protected website for regulator review.

Other Projects

e The preliminary data from the analysis of aliquots from fish tissue samples collected in early April from the
KIF Stilling Pond were received from Oak Ridge National Laboratory and shipped to Pace Laboratories in
Green Bay, WS for metals analyses. The data will be posted to the regulator web site the week of May 25.

FINAL REMEDIATION

Engineering (Dredge Cell)

e Received comments from TDEC on the dredge cell test. Concerns were raised about bringing ash into the
failed cell for the purpose of the test. The project team is working on a response. Efforts in completing
instrumentation and the underlying drainage layer have ceased.

e Began the process of collecting/developing information to support upcoming data quality objective

workshops to scope out the characterization effort that may be required to support the engineering
evaluation/cost analysis. Have developed a draft human health conceptual site model.

SAFETY

e 37 site specific HSE orientations conducted for the week, 280 to date.

o 2 Safety Observation Reports submitted for the week; 45 to date; 21 of which have been closed.

e |t was noted that the road surface on Dike C appeared to be slightly damaged as a result of heavy vehicles
trying to pass the bulldozer that is parked along this area. The bulldozer provides an anchor point for the
dredge in the stilling pond. The damage was investigated and subsequently repaired. Personnel were
notified to avoid driving heavy equipment on Dike C until the bulldozer is out of this area.

e Submitted draft safety and health plan to EPA on May 22"

COMMUNICATIONS

Stakeholders

e Attended Long Range Recovery Committee, reported on site progress, introduced presentation by ORAU
on health screening and introduced Stephanie Brown, EPA Community Involvement Coordinator.

o Met with local elected officials regarding property policies.
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e Conducted site tour for ORNL and University of the Pacific staff.

Communications
¢ Developing communications strategy and materials for Root Cause Analysis.

e Developing communications strategy and materials for selection of offsite disposal contractors.

o Posted to TVA’s Kingston Webpage: Updates on Water and Air Sampling.

Outreach Events
e Sent interim housing residents notice of lease terminations.

e Continued community interviews in preparation for the CIP (35 interviews completed).

e Continued appraisals, offers and closings on properties TVA is willing to purchase.

¢ Responded to numerous Congressional Inquiries and TVAINFO letters about residents.
e Fielded calls and visits from residents about various issues.

RIVER OPERATIONS

Weekly Average <
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Elevations and Flows cd?"
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(7 day average of data through May 24)
Emory River

Kingston Fossil Plant Flow: 790 cfs
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Melton Hill
low: 7,674 cfs
ev: 793.67 ft

Highway 58
Bridge

Average Kingston Velocity
0.547 ft/sec

Fort Loudoun
Flow: 20,695 cfs
Elev: 812.77 ft

Flow: 32,308 cfs
Elev: 740.95 ft

MEETINGS/CONFERENCE CALLS

Date | Time Subject

05/26 | 1300 Weekly Dredging Progress Meeting (ICP Conference Room)

05/26 | 1400 Dike Stability Meeting (ICP Conference Room)

05/27 | 0900 KIF Recovery Staff Meeting (ICP Conference Room)

05/27 | 1300 Construction Alignment Meeting (ICP Conference Room)

05/28 | 0900 Dredge Cell Test Team Meeting (ICP Conference Room)

05/28 | 1400 Rail Spur Meeting (ICP Conference Room)

05/28 | 1500 Safety Barricade/Rail Spur (Team Room)
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Kingston Recovery Project

Date:

May 18 — 24, 2009

Field Report

Safety Status/Personnel Casualty Summary

Since Last Report Adjustments To Previous Op Period Total Involved
First Aid 0 0 8
Recordable Incidents 0 0 2
Lost/Restricted Time Incident 0 0 0
Cenosphere/Debris Removal Quantity
Recovered Materials* Weekly Total Cumulative Total
Outreach Light Debris Removal — Land (bags) 0 18,867
Divisions A, B,C,D Cleanup Division shoreline cleanup (bags) 5,793 58,607
Total Tons of Light Debris 0 859.98
Cenosphere Slurry (gallons) 303,000 5,236,500

Environmental

Org Matrix Weekly Samples Total # of Samples Sample results

Collected Collected to date received verified
TVA Air (Real-Time)** 2,108 44,532 44,532 44,532
TVA Air (Analytical)** 15 1,386 1,276 1,039
TVA Surface/Utility Water 59 1,407 1,330 198
TVA Ground Water (spring and well) 0 5 5 2
TVA Ash 0 81 92 50
TVA Soil/Sediment 0 63 124 9

i Cumulative
Dredging Weekly Total Total
River Estimated Cubic Yards 17,900 156,000
Ash Pond 0 45,000
Stilling Pond Estimated Cubic Yards 3,100 3,100
Comments:
*May 18-22, 2009.
**May 18-21, 2009.
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